Author's response to reviews

Title: Essential role of miR-200c in regulating self-renewal of breast cancer stem cells and their counterparts of mammary epithelium

Authors:

Jian-Guo Sun (sunjg09@aliyun.com)
Zhong-Ming Feng (22346720@qq.com)
Xie-Wan Chen (893304871@qq.com)
Jun Qiu (zlpiyo@126.com)
Rong-Xia Liao (zlpiyo@aliyun.com)
Xing-Yun Liao (776325362@qq.com)
Xu Chen (276423923@qq.com)
Yan Li (407390154@qq.com)
Lu-Ping Zhang (284196729@qq.com)
Zheng-Tang Chen (314475914@qq.com)

Version: 7 Date: 1 August 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for your e-mail informing us that our manuscript (MS: 2064446519150315) is suitable for publication in the journal. Here, we have carefully studied the requests and made the appropriate revisions to our manuscript. The following are our specific responses:

1) Availability of Supporting Data section with details of microarray data deposition and accession numbers.

We encourage authors whose supporting data are available in an open access repository to include an "Availability of supporting data" section in their manuscript, before the Competing interests and Authors' contributions. The section should state the name of the repository in which your data is deposited and include a link to the dataset DOI. If all the supporting data are included as additional files the section should state this. If your manuscript has any supporting sequence data, microarray data, or proteomic data this must be deposited in the appropriate repository and a link to the dataset should be included in the Availability of supporting data section.

We have added the “Availability of supporting data” section to page 13 of the manuscript as required.

2) Email address of all authors on title page.

We have added the email address of all authors on title page.

I hope that we have adequately addressed the constructive comments. Besides, we have corrected some typos in the manuscript. We greatly appreciate your consideration of this manuscript for publication in your journal and look forward to hearing from you soon.

Sincerely,
Jianguo Sun, PhD